



# Article Investigation on Human Development Needs, Challenges, and Drivers for Transition to Sustainable Development: The Case of Qatar

Btool H. Mohamed <sup>1</sup>, Mustafa Disli <sup>2,\*</sup>, Mohammed bin Saleh Al-Sada <sup>3</sup> and Muammer Koç <sup>1,\*</sup>

- <sup>1</sup> Division of Sustainable Development, College of Science and Engineering, Hamad Bin Khalifa University, Qatar Foundation, Education City, Doha 34110, Qatar; btomohamed@hbku.edu.qa
- <sup>2</sup> Islamic Economy and Finance Group, College of Islamic Studies, Hamad Bin Khalifa University, Qatar Foundation, Education City, Doha 34110, Qatar
- <sup>3</sup> Department of Mechanical Engineering, Texas A&M University, Qatar Foundation, Education City, Doha 23874, Qatar; m.al-sada@qatar.tamu.edu
- \* Correspondence: mdisli@hbku.edu.qa (M.D.); mkoc@hbku.edu.qa (M.K.)

Abstract: Human development has been widely accepted as both the primary goal and driver of sustainable development. Human development is critically important for countries attempting a radical transformation from a resource-based economy to knowledge-based, sustainable development, like Qatar. This paper examines Qatar's human development needs and challenges with respect to its economic, demographic, and educational development goals. Following a comprehensive literature review, domain-specific documentary information and data were collected and analyzed, along with qualitative insights and data obtained through semi-structured interviews with local professionals from various sectors, to gather opinions about the needs, challenges, drivers, and recommendations for human development planning are proposed as a guide, for Qatar and similar countries, for transitioning from a resource-based economy to knowledge-based sustainable development. This framework and the understanding of the needs, challenges, and drivers based on the qualitative research, in conjunction and in comparison with the literature analysis, are expected to assist decision-makers in devising, and more importantly implementing, tailored policies and roadmaps focused on four key dimensions: skills (education), citizenship, leadership, and health.

**Keywords:** sustainable development; human development; human capital; human capacity; Qatar; resource-based economy; knowledge economy

# 1. Introduction

Human development (HD) has been recognized as the ultimate goal and main driver of sustainable development, which strives for delicately balanced economic, social, and environmental outcomes [1]. This recognition, however, is not without challenges and resistance. For decades, labor has been misrepresented as the sole factor of production in economic growth, relegating social and environmental development to the background [2]. Increasing emphasis on knowledge-based discourses for solving sustainability challenges has required countries to reconsider and rejuvenate their development frameworks [3]. This includes many countries that rely on natural resources but aim to achieve sustainability.

Although the hydrocarbon sector continues to drive economic prosperity in the Gulf Cooperation Council (GCC), questions concerning the sustainability of resource-reliant models have arisen due to systemic shocks, resource depletion, and changing demographics and consumer habits [4]. The policy challenge of how Gulf countries can achieve sustainable development and move away from dependence on extractive industries represents a topic of rigorous scholarly debate over the past decades [5]. It has long been postulated that



Citation: Mohamed, B.H.; Disli, M.; Al-Sada, M.b.S.; Koç, M. Investigation on Human Development Needs, Challenges, and Drivers for Transition to Sustainable Development: The Case of Qatar. *Sustainability* **2022**, *14*, 3705. https://doi.org/10.3390/ su14063705

Academic Editor: Carla Maria Marques Curado

Received: 30 December 2021 Accepted: 11 March 2022 Published: 21 March 2022

**Publisher's Note:** MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



**Copyright:** © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). using the resource advantage significantly boosts countries' fiscal revenues and economic growth [6]. However, these models have been increasingly challenged, as GCC member states do not effectively use resource revenues to follow a path of sustainable growth [4]. Furthermore, overreliance on natural resources to drive economic growth produces negative externalities such as weak institutional and legal frameworks and underdeveloped non-extractive industries [7,8] confirms this, demonstrating that natural resources crowd out human capital and decelerate economic development. With the rest of the world increasingly embracing new technologies as a source of long-term growth, it is evident that the path of the Gulf States to sustainable development and diversification requires a refocus from the hydrocarbon-led economic model to those areas considered strategic in the development of a knowledge-based economy.

The challenges facing a transition to sustainability have also been acknowledged by the state of Qatar, which recently laid out grand policy strategies under the umbrella of the Qatar National Vision 2030 (QNV2030) [9]. With its emphasis on a knowledge-based transition, QNV2030 involves reconciling conflicting policy priorities to achieve sustainable development. It furthermore addresses the population imbalance between local citizens and expatriates, and the country's aspirations to maintain its current per capita GDP in the coming decades. The Qatari human development plan aims to prepare its citizens for knowledge-intensive industries and ensure the steady replacement of low-skilled expatriate workers with well-trained foreigners [10].

Qatar's challenge in transitioning to a knowledge-based economy merits further research, as a failure to move from a resource-based to a knowledge-based economy will inevitably yield long-term ramifications for the country's development. Countries, particularly those suffering a knowledge deficit, have much to gain from moving to the "knowledge society" as returns on knowledge acquisition increase [11]. According to [12], government effectiveness, education spending, and academic publications positively influence economic development. Moreover, education represents a powerful tool against fraudulent reporting [13] and corruption [14]. Creating a productive society is closely linked with knowledge acquisition and economic demand. The relationship between higher-value-added activities and knowledge-intensive activities is increasingly prominent. Besides comprising the foundation of competitiveness worldwide, these activities will eventually generate the wealth of the future; hence, they are crucial for the development of countries in transition. The state plays a vital role, primarily through research and education, which are areas that do not result in immediate profits but are ultimately essential to any knowledge acquisition system [15]. Ultimately, how dynamically a society participates in knowledge acquisition and how effectively it contributes to human development depend on its social, cultural, economic, and political structures.

Qatar aspires to transform itself into a sustainable knowledge economy in the coming decades and gradually move away from its heavy dependence on oil and gas resources [16]. Thus, understanding the status and recent trajectory of the country's demographics, economic structure, and workforce composition is critical in determining if this vision is achievable. This is particularly true given that sustainable development can be realized only with tailored strategies and adequate preparation of human development [17,18]. Thus, the question underlying this study is how to formulate policies for educated, talented, and forward-looking human development that will identify a trajectory toward sustainable development based on an innovation-driven knowledge economy.

Some scholars have been more inclined to examine opportunities for human development in light of the 2022 World Cup, which Qatar will host [19]. Although such scholarly efforts are needed to assess and appraise Qatar's short-term development plans, it should be recognized that mega-events rarely bring sustainable economic growth to host countries [20]. Sustainable job creation was a major selling point in attracting previous similar mega-events, but expectations surrounding these have rarely been satisfied [21,22].

Other research compares management of human resources within the Gulf by exploring how different GCC states have addressed their policy challenges [4,21]. Still, such investigations lack the in-depth assessment of individual countries' responses and seem to generalize policy prescriptions that assume homogeneity across the Gulf States. Similarly, *Qatarization* and its impact on the Qatari labor market have also been studied [22,23] (*Qatarization* is a focus of QNV2030 and aims to increase the number of Qatari citizens employed in both the private and public sectors). Yet, these investigations have focused on the systemic challenges created by policy without recognizing that the path toward a sustainable knowledge economy requires a complete transformation of the Qatari education system.

The transition from a resource-based economy to a knowledge-based economy is of contemporary relevance and merits an in-depth, country-specific inquiry [24,25]. This study aims to develop a comprehensive framework to examine and assess the human development strategies of Qatar by exploring the measures introduced in various areas (including education, the labor market, and *Qatarization*). It furthermore takes the systemic social and economic challenges that the country faces into context. To achieve this, two research questions (RQs) are explored:

**Research Question 1 (RQ1).** Which gaps, trends, and needs exist in Qatar's human capacity to contribute to the goals of QNV2030, considering Qatar's economic diversification and sustainable development options?

**Research Question 2 (RQ2).** Which progressive and adaptive policies, strategies, and roadmaps are needed to contribute to the goals of QNV2030, considering the national, regional, and international environment, trends, and challenges?

The rest of this paper is organized as follows. Section 2 discusses Qatar's case in detail, including a comparative analysis of Qatar's education, demographics, and human development against benchmarked countries. This section is built on secondary data and intends to answer RQ1 of this study. Section 3 describes the methodology of the interview process, along with sample selection. Section 4 presents findings of the qualitative study and attempts to answer both RQ1 and RQ2. Section 5 presents a synthesis of overall findings from the literature review and qualitative research. Section 6 integrates secondary data analysis with stakeholder feedback to develop a framework for Qatar's human development strategy, thus completing the analysis of RQ2. Finally, Section 7 summarizes the research outcomes and provides a critical reflection on the limitations of this study as well as possible avenues for future research.

# 2. Literature Review

#### 2.1. Status of Human Development in Qatar

Perspectives on the place and characteristics of humans as a component of the economic system have considerably changed in the transition from resource-led growth models to knowledge-based economies and sustainable development. During the 19th and 20th centuries, humans were considered alongside other capitals of land and labor as a production factor for economic growth [26]. However, with the automation and technological revolution, as well as the changing approach of capitalism in the 20th century, the perceived role of the human has evolved to include contributing capacity for enabling economic development [27]. Later, at the beginning of the 21st century, the innovation-driven, knowledge-based sustainable development paradigm and a heightened focus on society and environment [28] led to recognition of humans as enablers and building blocks of sustainable development [27].



Figure 1. Human capital theory [27].

Human development is a multi-dimensional process that combines various forms of social, economic, and political progress to enable innovation and the development of human potential [1]. As a result of human development, this enhancement in human potential is a necessary requirement for a country's transition to a knowledge-based economy and thus sustainable development. Knowledge economies require research-based innovation and development on which to capitalize [29]. These are intangible assets that a country can produce via dedicated focus on particular areas, such as education and workforce development. It is also important to work on demographics to make them more sustainable and the country's workforce more efficient. Therefore, to transition to a knowledge economy, countries must channel their economic preferences, with the overall purpose of building human potential. Based on secondary data, this section attempts to answer RQ1 by examining Qatar's current economic, demographic, and educational trends. We furthermore compare Qatar's performance against benchmarked countries using the Education pillar of the Human Development Index (HDI) and the Human Capital and Research sub-index of the Global Innovation Index.

### 2.1.1. Qatar's Economic Trends

Qatar is one of the wealthiest countries in the world, but its economy is criticized for its high dependency on oil, gas, and the derivatives of these. Qatar 2030's vision is to build a society that promotes equality, justice, and goodwill [30]. Four fundamental developmental pillars (human, social, economic, and environmental) are identified in the policy [30,31]. Some elements of this system are already integrated into current policies, some are under construction, and others are yet to be implemented.

Qatar has experienced fast-paced economic development in recent decades owing to oil and natural gas revenues. Natural gas production has increased sharply, from 2.2 million tons per year in 1997 to 92 million tons in 2013, making Qatar the world's largest exporter of liquefied natural gas (LNG) [32]. A third of global LNG trade can be attributed to Qatari exports, and 80% of Qatar's exports are oil and gas products [33]. Qatar's economic and financial policies have thus been driven by factors connected to hydrocarbon production and export. However, Qatar has also made significant strides in diversifying its economy, investing in non-oil sectors. Qatar aims to transform into a knowledge-based economy and society, reducing the contribution of oil and gas to its economy. Investments in the financial sector at both national and international levels have started to yield dividends, increasing Qatar's diversification compared to its neighbors [34]. Despite this, diversification initiatives have yet to show their actual, long-term impacts on Qatar's overall economic outlook. As mentioned, the economy still relies heavily on natural resources, followed by the services and manufacturing sectors. These sectors have experienced further growth thanks to Qatar's hosting of the 2022 FIFA World Cup, and sectors like construction and services have also started to grow. As current projects conclude, the question becomes which sectors, other than oil and gas, Qatar should focus on to become a sustainable knowledge economy [35].

This study will thus assess Qatar's hydrocarbon and construction sectors as the dominant contributors to GDP and the subsequent impact on a sustainable development vision.

#### 2.1.2. Demographic Trends

Qatar's population has increased drastically in recent years, especially since 2010, primarily because of the influx of low-skilled workers for construction projects related to the FIFA World Cup. Additionally, the gender distribution has shifted toward a male majority since 2010, as the numerous construction and industrial projects have attracted almost exclusively male immigrants. Since 2010, there has been a surge in the population of 25 to 35-year-old males, yielding a significantly high ratio of men to women—the highest in the world. Once World Cup construction projects are completed, assisting Qatar's change to a knowledge economy and carefully managing the *Qatarization* process will require that immigration policies are changed to attract highly-skilled ex-pats. Regarding the general population, Qatar is a country with a small proportion of native citizens embedded in a larger expatriate population. Data from the official Planning and Statistics Authority (PSA) show approximately 333,000 Qatari nationals living in Qatar as of early 2019, meaning that Qatari nationals accounted for only 11.7% of the population in mid-2019 [36]. As the number of migrants in Qatar continues to grow, the local population becomes an even smaller minority, with natural population growth unable to keep up with the growing number of expatriates pouring into the country.

A knowledge-based society efficiently produces, diffuses, and uses a wide range of knowledge. Thus, a well-educated and skilled population is required to transition to a knowledge society, as shown in multiple reports from the World Bank and the Organization for Economic Co-operation and Development detailing both successful and futile transformations [37]. To expand its skilled workforce, Qatar must attract skilled migrants. One way to do this is by focusing on international students in essential fields. Encouraging the mobility of international students is beneficial for both home and host societies, as student migration can positively influence economic growth for both sending and receiving countries [38]. However, migrants themselves must reap most of the gains through higher earnings for the process to succeed.

Qatar must also pay close attention to developing tailored policies to overcome the shortcomings presented by the skilled, well-educated population required for the transition to a knowledge economy, which fall under the assessment of this study.

#### 2.1.3. Qatar's Education System

Any country wishing to transform into a knowledge economy must establish the six fundamental pillars: leadership, human capital, innovative capabilities, IT infrastructure, financial resources, and innovative climate [39]. As depicted in Figure 2, the native Qatari population is reasonably well-educated, with most males educated up to the secondary level, though Qatari females have a greater inclination to pursue higher education. Conversely, Qatar's expatriates have less robust education portfolios, with most men educated only at the primary level. The main reason for this less-educated, male-dominated non-Qatari population is that men are typically involved with hard-labor jobs in Qatar's large infrastructure enhancement projects. When Qatar's investments in the World Cup end, a large percentage of these non-Qatari laborers, who have only contractual appointments are



expected to return to their native countries, affecting the demographics of Qatar and the education spectrum.

**Figure 2.** Population distribution by nationality and education level. Source: adapted from [40]; total refers to population (left *Y*-axis).

More investment and focus have been placed on Qatar's education field, and since 2002, comprehensive institutional reforms have aimed to strengthen the education and schooling system [41], a process that began with the reform of the K-12 education system. In 2003, the Supreme Education Council (SEC) was established to oversee the deployment of subjects/fields offered in the independent school system and develop the accountability of the system. In 2016, the SEC was reformed into the Ministry of Education & Higher Education [42]. By 2011, a bilingual Arabic/English system had replaced the solely English teaching system for mathematics and science [42]. Under the umbrella of the Qatar Foundation (QF), the country has invested in numerous education, research, development, and innovation initiatives to support the knowledge economy by attracting international branch campuses to Education City and instituting the Qatar National Research Fund (QNRF), the Hamad Bin Khalifa University (HBKU), Research Institutes (QEERI, QBRI, QCRI), the Qatar Science and Technology Park (QSTP), and Qatar Academy schools. Since 2000, Qatar has made remarkable investments in the education and health sectors. The number of schools, classrooms, IT infrastructure, et cetera, increased alongside the population because of economic growth, which was fueled mainly by natural gas revenues. The numbers of students and teachers have also increased significantly, and there is a greater variety of school/curriculum types thanks to liberal policies implemented to accommodate the wide variety of expatriates arriving. However, despite the variety of schools and universities, the question is whether the education imparted in the fields preferred by the population is conducive to building a knowledge economy.

Expenditures assigned for education comprise another indicator of Qatar's educational progress. Following QNV2030, the Qatari leadership has assigned a larger share of GDP to education. The government has invested massively in infrastructure enhancements to improve the quality and quantity of educational institutions at the primary, secondary, and tertiary levels. In addition to education expenditure, student–teacher ratio represents another critical aspect of developing a knowledge economy [43]. Current figures are very

encouraging; the average number of students per teacher is 15 at the pre-primary level and drops to 9 at the secondary level, increasing the quality and duration of student–teacher engagement.

The Qatari enrollment rate has been stable while the non-Qatari enrollment rate has increased, commensurate with the increase in population. However, the enrollment of the Qatari population in tertiary education has been increasing, while non-Qatari enrollment has been stable. This means that tertiary enrollment figures do not follow the trend observed in secondary enrollment rates, suggesting that non-Qataris prefer attending international universities outside of Qatar, which raises the question of how Qatar can attract and retain human capital. Another feature is the higher enrollment rate of Qatari women in tertiary education than men, as shown in Figure 3. One reason for this is the cultural constraint on women, which restricts them from gaining international experience, yet there are more Qatari women in tertiary education in international universities in Qatar than Qatari men. Another reason is that Qatari men may lack an incentive for higher education, as they are offered lucrative positions in the public sector after graduating from high school [44].



Figure 3. Gross secondary and tertiary enrollment rates in Qatar (2011–2019), adapted from [45].

In light of the observations above, it is important to note that some majors pair particularly well with STEM (science, technology, engineering, and mathematics) areas. However, the fields chosen by Qatari students do not necessarily align with these majors. This observation contradicts both knowledge economy requirements and the aims of *Qatarization* policies. Figure 4 shows the number of graduates from various disciplines for both the Qatari and non-Qatari segments of the population between 2013 and 2019.



**Figure 4.** Graduates of public colleges and universities based on specialization and nationality, adapted from [40].

The Qatari education system has been subject to significant modifications and improvements over the past decade. However, benchmark indicators still show that academic achievements of Qatari students are behind those of their counterparts in the developed world [42]. According to research commissioned by PISA (Program for International Student Assessment) [46], examination results showed that Qatar ranked 61st out of 65 studied countries regarding quality of education. Only students from Panama, Peru, and Kirgizstan scored lower than those from Qatar [47]. The education chasm between the performance of Qatari students and students from more advanced knowledge-based economies is readily apparent [48]. Test results further showed that Qatar's average score in reading literacy was 407 points, while the average across Organisation for Economic Co-operation and Development (OECD) countries was 487 points [37]. The same trend is observed in mathematics, with a shortfall of 70 points compared to other countries' scores [37]. A more telling indicator is the number of students scoring below proficiency thresholds. Nearly 50% of Qatari students appeared to have only basic proficiency in reading, a score equivalent to that of Saudi Arabia, and the number of students with only a basic reading level was nearly five times higher than that of the knowledge-based economies of Finland, Singapore, and China [34].

Some have argued that these discrepancies are a consequence of long-standing reforms to education [41,48,49], yet, it is noteworthy that Qatar underperforms in educational outcomes. Despite improvements in the education sector, PISA test results suggest that substantial transformation in the education sphere is necessary for the country to become a knowledge-based economy.

Such results further suggest that Qatar's economic and educational reforms have not produced the desired level of change. There is a need for systemic improvements in the education sector to ensure that best practices for developing human potential are employed and that the country's education sector can learn from the experiences of other countries [50].

#### 2.2. Comparative Analysis of Qatar and Benchmark Countries

Different countries design and follow different strategies for transforming to knowledge-based sustainable development. In this section, Qatar is compared against a group of benchmark countries using the Human Development Index (HDI) [51] and Global Innovation Index (GII) [52]. The HDI index is measured through three basic dimensions, including long healthy life (health), access to knowledge (education), and decent standard of living (employment). The GII index has two sub-indexes: innovation input and innovation output. Innovation input incorporates the scores for all enablers: (1) institutions, (2) human capital and research, (3) infrastructure, (4) market sophistication, and (5) business sophistication. The scores for (6) knowledge and technology outputs and (7) creative outputs are combined in the innovation outputs' sub-index. This study focuses only on the access to knowledge (education) sub-index of HDI since education is considered to be the most transformative tool for sustainable development [51]. The same reasoning holds for the human capital and research sub-index of GII. Therefore, we use the Education component of HDI and Human Capital and Research sub-index of GII as each other's complements. The study in [52] proposes a tight connection between sustainable development and quality of education, since education necessitates adequate conceptualization, given the importance of education and human development in creating sustainable, long-run development; education further ensures proper implementation of policies proper implementation of policies and recommendations for measurable changes, as well as the tools needed to monitor and evaluate this change.

The five countries selected for comparative analysis were Norway, Singapore, South Korea, Bahrain, and the United Arab Emirates (UAE). South Korea, Norway, and Singapore were selected as benchmark countries on the grounds of their similarities to and differences with Qatar. South Korea presents a good example of a country that has undergone a successful transformation from an agriculture-based economy to a knowledge-based economy thanks to its strategic, progressive planning and its leadership, which has been wise, dynamic, and long-term [53]. Furthermore, the careful implementation of strategies supported by continuous monitoring and assessment is a prerequisite for South Korea's successful shift to a knowledge-based economy. Qatar has adopted a similar long-term strategic approach, as evidenced by the introduction of QNV2030. Like Qatar aims to do, Norway transformed from an oil-centered economy to a knowledge economy. Norway offers an example of transformation based on participative decision-making and a well-coordinated strategy between economic and education/workforce development, from which researchers and practitioners can learn [54]. Norway's successful immigration policy is another point Qatar should consider in shifting from its prevailing low-skilled worker structure to policies attracting more skilled and semi-skilled workers [55].

Singapore is of similar size and demographics to Qatar, with a relatively small population and land that is limited but holds strategic geography. Singapore is also known for its diversified population and openness for selective naturalization and immigration, as well as long-term leadership guiding its transition to a knowledge-based economy [56]. Singapore's experience additionally indicates that improving education quality, which creates an innovation-driven environment, and attracting skilled labor are essential in creating more sustainable economic growth [49]. Bahrain and the UAE were selected due to similar populations and richness in natural resources, as well as geographical and cultural proximity. Their national visions are, furthermore, similar to QNV 2030, with a knowledge-driven aspiration.

In 2020, Qatar ranked 45th out of 189 countries based on its HDI of 0.848, up from 0.750 in 1990 [57]. Qatar ranks reasonably well overall but still lags behind benchmarking countries. Since education is the primary focus of this study, we conducted further investigation of the HDI sub-index concerning education. In determining Education Index, two dimensions are considered: (1) mean years of schooling and (2) expected years of schooling. Table 1 shows that Qatar performs poorly in terms of education or knowledge output. Thus, greater emphasis must be placed on education, an identified gap highlighted in this study.

Country	Rank & (HDI Value)	Education Index	Expected Years of Schooling	Mean Years of Schooling
Norway	1 (0.957)	0.93	18.1	12.9
Singapore	11 (0.938)	0.844	16.4	11.6
South Korea	23 (0.916)	0.865	16.5	12.2
United Arab Emirates	31 (0.98)	0.802	14.3	12.1
Bahrain	42 (0.852)	0.769	16.3	9.5
Qatar	45 (0.848)	0.659	12	9.7

**Table 1.** Qatar compared to benchmark countries via Human Development Index (HDI) and Education Index 2020 [57–60].

Table 1 shows that Qatar ranks at 45 with an HDI value of 0.848, which is the lowest of the six countries considered. Norway demonstrates the highest HDI value, 0.957. From the Education Index, Norway clearly outperforms the five other countries, including Qatar, which has the lowest Education Index, which then decreases overall HDI for Qatar. Hence, if Qatar improves its education (as measured via Education Index), a higher HDI value and ranking will result.

Similar results are obtained when comparing GII Human Capital and Research components between Qatar and the benchmark countries. From Table 2, we can observe that poor performance is attributed mainly to the subcategories of education and R&D. Qatar's performance in the education subcategory is explained by the small percentage of government expenditure to overall GDP. This ratio is, however, substantially influenced by oil price fluctuations. However, the weak results from the PISA test, which indicates quality of education, suggest that the government must make reforms to improve the quality of the education system. If such reforms are not enacted, supplying Qatari nationals for the local private labor market may remain challenging [61]. Qatar also does not perform well in R&D because of the limited number of researchers within the population—577.3 researchers per million inhabitants [62]. This underscores the importance of nurturing high-caliber universities in Qatar so that local talent can flourish to address domestic needs and serve as ambassadors for Qatar on global platforms [63].

Although Qatar has prioritized education, human capital development, and R&D as part of its QNV2030 since the 2000s, these findings from recent HDI and GII comparisons are alarming.

#### 2.3. Identified Gaps

The literature review identified numerous gaps after studying three thematic areas (demographic trends, Qatar's education system, and economic trends). First, studies of demographic trends show the large influx of low-skilled workers as a key challenge that continues to affect the country's current demographic structure, which then affects its overall performance, efficiency, and sophistication in human development, education, and economy. This area of research is not complete, however, as a lack of understanding regarding the immigration policies necessary to develop a knowledge-based economy exists. Second, studies on economic development provide sufficient information regarding the economic effects of over-reliance on gas and oil. There is, however, a lack of clarity on the high-tech, knowledge-intensive, and green sectors in which the country must invest to establish a sustainable economy. Studies investigating the education system indicate that Qatar has made significant progress in education as a result of investments in infrastructure and numerous initiatives related to education, research, development, and innovation. The gap identified in these studies is that current practices do not align with the government's aspiration to achieve the sustainable development goals outlined in its QNV2030.

Country	Ranking of Human Capital and Research (Index Value)	Ranking of Education	Ranking of Tertiary Education	Ranking of Research & Development (R&D)
Singapore	9	54	2	15
	(58.7)	(54.0)	(63.1)	(59.1)
South Korea	1	22	13	1
	(67.4)	(61.5)	(51.0)	(89.8)
Norway	13	3	42	19
	(56.8)	(75.3)	(39.7)	(55.5)
United Arab Emirates	22	61	3	28
	(49.9)	(52.0)	(59.2)	(38.6)
Bahrain	83	81	73	82
	(26.3)	(44.1)	(30.5)	(4.2)
Qatar	75	94	37	67
	(29.8)	(40.1)	(42.0)	(7.4)

**Table 2.** Qatar compared to benchmark countries through GII human capital and research components (GII) [62].

The literature review in Section 2 attempts to clarify the gaps, trends, and needs for Qatar's sustainable development ambitions. The following sections will supplement the analysis thus far by exploring the experiences, views, and insights of interview participants. Furthermore, the qualitative study will shed light on what policies must contribute toward sustainable development.

### 3. Methodology

#### 3.1. Design Thinking Approach

As shown in Figure 5, the design thinking approach involves five phases: discovery, interpretation, ideation, experimentation, and evolution [64]. In the discovery phase, a comparative examination of Qatar's demographic, economic, and human development trends and educational issues was achieved by performing a comprehensive and comparative literature review. Outcomes of the discovery phase comprised the introduction and literature review sections of this manuscript.

Second, during the interpretation phase, our team identified gaps and composed a problem statement for Qatar's hydrocarbon and construction sectors, as they represent the main economic contributors to GDP. This domination impacts human capital needs in terms of demographics and skills, alongside the weakness of the industry–university partnership in terms of education, training, and research. This phase demonstrates the serious and salient shortcomings that must be recognized and addressed with appropriate, effective, and long-term policies and recommendations, with contribution from all stakeholders.

Third, in the ideation phase, we conducted semi-structured interviews with stakeholders from various sectors, including government, private, and education sectors, for two main purposes: (1) to identify any further shortcomings, needs, and challenges that exist in Qatar's human capacity to contribute to the goals of QNV2030 and thus sustainable development, and (2) to propose a framework for progressive and adaptive policies needed to achieve the excellence in education, training, and research for meeting the needs of economic diversification and sustainable development. This phase thus yielded the methodology and results sections. All findings were then refined by analyzing them into themes and subthemes using thematic analysis, after which a set of initial recommendations was generated. Finally, we wrote discussions, recommendations, and conclusions of the study.



**Figure 5.** Design thinking process followed by this study. \* Indicates that the associated section is written in the published paper [27].

The fourth and fifth phases are beyond the scope of this study, requiring a long, tedious, and complex implementation procedure in a real-world environment as well as an impact study on both the society and economy. Hence, this manuscript concludes with recommendations derived in the third phase and references the remaining phases as future work. An initial step of the experimentation phase was conducted, which involved proposed policy validation, and was conducted through a second round of interviews and published in another study [27].

#### 3.2. Interviews

A qualitative research interview is a structured discussion in which the interviewer solicits information and opinions from participants about a particular subject [45,65]. For this study, the purpose of conducting semi-structured interviews was to obtain perspectives from participants from the public, private, and academic sectors to (1) confirm the existence of the identified economic, demographic, and educational gaps regarding sustainable development, (2) categorize these gaps into major themes based on interviewee opinions, (3) identify any further gaps and challenges in response to RQ1, and finally, (4) propose recommendations in response to RQ2 that drive the quality of Qatar's human development toward sustainable development. We used semi-structured interviews for data collection, a procedure that establishes some structure through open-ended questions and allows interviewees enough flexibility to provide new insights [61,66]. Furthermore, semi-structured interviews suit this topic of investigation because they provide in-depth insights into human development challenges that Qatar faces and highlight a roadmap toward sustainable human capital development. Additionally, this format allows for follow-up questions and clarifications as deemed necessary [67,68]. Research was conducted between February 2021 and April 2021. Fifteen open-ended interview questions were designed to elicit further details pertaining to (i) the status of human capital development and (ii) Qatar's strategies to develop human capacity (see Appendix A). These open-ended questions were derived from (i) previous literature about human development, (ii) gray literature about Qatar, of which a major portion is incorporated in the previous section (such as the QNV2030 document, Qatar's national development strategies, and annual reports from Qatar's Ministry of Development and Planning and its Educational Ministry), and (iii) the purposes of this study. Each interview lasted between 30 and 45 min, and participant responses were

recorded in a "data notebook" to maintain confidentiality. Before the survey, a version of this questionnaire was pilot-tested with six colleagues (pre- and post-doctoral students and faculty with relevant research interests) and two potential study participants. These pilot interviews did not result in substantial modifications to the questions but were nonetheless helpful in refining them and in developing follow-up inquiries.

### 3.2.1. Sampling and Participant Selection

Using purposive sampling, we assembled 20 participants with a balanced gender distribution from a wide variety of backgrounds and qualifications (from the private, public, and academic sectors) to capture and integrate multiple perspectives to achieve a more advanced, holistic view of human development. The general criteria for selecting interviewees were that they had professional and relevant experience and were willing to offer perspectives on Qatar's attempts to transform its economy into a knowledge economy, and the country's human development needs and issues. More specifically, we identified people with sufficient knowledge and experience pertaining to human development practice, including policymakers, human resource practitioners from the private and public sectors, and academics and teachers with relevant disciplinary expertise. Using a purposive sampling procedure allowed us to gain an in-depth understanding of interviewees [69]. Besides purposive sampling, snowball sampling was conducted by asking those contacted to recommend others within or outside their organizations who might provide useful insights. Some participants had a great deal of experience and represented decision-makers in their fields; almost half also had experience in two or more sectors, adding valuable depth to the study. Informed consent was given by all study participants. Table 3 presents the demographics of interviewees.

Participant	Gender	Age	Qatari	Profession	Education
1	М	50s	No	Economic consultant	Ph.D.
2	М	50s	Yes	University president	Ph.D.
3	М	40s	Yes	Engineering director	Ph.D.
4	F	20s	Yes	Recruitment specialist	Master's
5	М	50s	No	University lecturer	Ph.D.
6	F	40s	No	Researcher	Ph.D.
7	М	30s	Yes	Assistant deputy minister	Master's
8	F	20s	Yes	Engineer	Bachelor's
9	М	60s	Yes	Former minister	Ph.D.
10	М	30s	Yes	Quality and planning	Bachelor's
11	F	30s	Yes	Engineer	Master's
12	М	40s	No	Secretary-general	Master's
13	F	50s	Yes	Former teacher/stay-at-home parent	Bachelor's
14	М	30s	Yes	Engineer	Master's
15	F	20s	Yes	Legal researcher	Bachelor's
16	F	20s	No	Lawyer	Bachelor's
17	М	30s	Yes	Manager	Bachelor's
18	F	30s	Yes	Engineer	Ph.D.
19	F	30s	Yes	Researcher	Master's
20	М	60s	No	Engineer	Bachelor's
Total:	20				

Table 3. Background and characteristics of respondents.

# 3.2.2. Thematic Analysis

Data from semi-structured interviews were transcribed and subjected to thematic analysis to identify significant elements related to topics arising from the current infrastructure in Qatar and the factors assisting or impeding the creation of a successful human development model suitable for Qatar. In conducting thematic analysis, we followed a six-step process recommended by [70]: (1) familiarization; (2) coding; (3) generation of themes; (4) reviewing themes; (5) defining and naming themes, and 6) writing up. Transcripts of the interviews were first read several times and manual coding was used to highlight the most frequent answers and categorize the responses. These responses were classified into subthemes, which were then interpreted into themes. The research team reviewed these themes and subthemes to ensure that all data related to the research questions were included; Figure 6 illustrates these themes. Each (sub)theme helped illuminate topics relevant to this study. In defining and naming (sub)themes, each (sub)theme's exact meaning was formulated and its relationship to the research questions determined. We report the thematic analysis findings in the next section. From interview transcripts, relevant phrases were identified [71] and gathered into "codes" that represent a collection of similar phrases sorted into the same place [72]. Over 500 codes were initially identified in interviewee responses, which were then combined into potential themes [70]. Codes were merged, eliminated, and moved during further review. There was no limit to code reduction, which was based on information saturation and yielded three main themes and seven subthemes that demonstrated relevance to the research questions.



Figure 6. Interrelations between research questions, main themes in the interviews, and findings.

3.2.3. Validation and Trustworthiness

The supervisory committee members reviewed the interview questions in terms of language and applicability to the research questions. Based on their comments, interview questions were modified and improved as an initial review. Following that, three pilot interviews were conducted with participants to capture their opinions and observe the flow of discussion. This offered an opportunity to detect possible flaws at an early stage. As previously mentioned, pilot interviews resulted in only minor modifications to interview questions. Clarifications were provided where needed to ensure that each participant interpreted the questions similarly. The results were presented to most interviewes and interested readers in a follow-up survey and validation interviews [27].

#### 4. Results and Discussion

As mentioned, the thematic analysis resulted in three main themes and seven subthemes, as shown in Figure 6.

### 4.1. Theme 1: Current Infrastructure

This theme is related to RQ1, and the responses focus primarily on the status of infrastructure in Qatar and its interaction with human capacity and sustainable development. They offer the opportunity to determine if Qatar possesses the required factors for developing its human capacity to support a knowledge economy. Under the theme of current infrastructure, interview responses contain two subthemes: QNV2030 and Facilities. Regarding QNV2030, most respondents saw it as a current driver of development. Respondents additionally noted that QNV2030 was developed to support high standards related to Sustainable Development Goals (SDGs) [73]. Their main concern was whether it was possible to achieve the targets set in QNV2030 by 2030. Participants also expressed concerns about the concept of citizenship; ideas of citizenship and belonging are poorly defined, and young people may not understand their importance in the development process. In particular, interviewees observed that the younger generation are not provided with the skill set to become effective community members. In addition, a sense of entitlement from being a Qatari citizen may not be ingrained in younger minds, leading to a lack of ambition and engagement among this population group. This absence of entitlement represents a concern not only on a personal level but on a societal level as well.

To develop the necessary human capacity through education, several institutions have been formed in Qatar. Such institutions were established to monitor and evaluate the performance of schools, teachers, and students, as well as support parental perspectives, as all of these factors are directly associated with human capital development [71,74]. However, interviewees from both the education sector and the broader economy considered the role of the evaluation authority in Qatar to be routine and repetitive, with no tangible outcomes or changes regarding the development of skills and talent.

The second identified subtheme concerns infrastructure and facilities. Most interviewees stated that Qatar has several supporting fundamentals required for human capital development. These include infrastructure items, such as transportation, energy, access to electricity and clean drinking water, a healthcare system, schools, and universities. These factors have a direct and positive impact on human development [74].

# 4.2. Theme 2: Affecting and Transformation Factors

In this section, we attempt to identify the affecting and transformation factors that influence the path of human development. These include demographic, social, cultural, and educational factors, as well as workplace environment. They are classified into the following subthemes: existing issues, challenges and barriers, supportive factors, and needs [75]. This section delves further into RQ1 and provides insights into RQ2. Concerning existing issues, demographics and current population structure were referenced by all interviewees, with particular emphasis on the current population's imbalance regarding gender, skills, and nationalities. Respondents also believed that, after the World Cup, the required skills for sustainable development must be examined and reviewed in terms of quantity and quality. Another existing issue is the conflict between educational projects and vision. A strong education system is a key pillar in human development [76]. However, interviewees saw a conflict between current educational projects and Qatar's desired human capacity and sustainable development. Primarily, respondents were concerned that Qatari men lack the incentive to pursue higher education. Another issue concerns the practice of the Ministry of Administrative Development, Labor and Social Affairs offering job contracts to university students during their studies, which discourages Qatari citizens from pursuing professional or academic ambitions. This practice seems to obstruct creativity, entrepreneurship, and innovation.

Furthermore, a high number of students graduating from secondary school are not qualified to enter university, with academic backgrounds that are too weak to meet the requirements of most universities. Many secondary schools have switched to Arabic as their language of tuition, yet both local and international universities use English as their primary instruction language [77]. Many students have thus lost their chance to pursue education because of language barriers, which is a major blow to the country's aim of increasing human capacity. Some interviewees referred to the policy of *Qatarization*, observing that if most Qataris studying in universities are assigned to jobs in the government, the private sector will remain un-Qatarized [78]. Additionally, some stated that collaboration between the education system and the workplace is weak. There are no clear requirements from the workplace regarding what majors are needed. This broken chain between supply and

demand has many negative influences, including on finances and student motivation. The second subtheme involves the challenges and barriers to achieving a knowledge economy. The opportunities in the public sector, mainly in the military, are attractive in terms of the privileges offered, making them likely to override other ambitions. Qatari men may thus question why they should pursue higher education when prestigious, secure positions are available to them at age 18. Another issue is the lack of role models. In addition, successful parents, who can be role models, are rarely highlighted in Qatar. Qatar depends heavily on expatriate teachers (around 95%). Most are women, who cannot necessarily be the best role models for Qatari boys [79]. A high number of Qatari female teachers likely went into teaching because cultural and social constraints made it their only suitable employment option [80]. Another significant challenge to human capital development is the loss of female talent [81]. Many educated Qatari women do not join the workforce, or they work in sectors different from their education major. Most interviewees considered this a significant issue and noted that its explanation is deeply rooted in local, social, and cultural aspects. Women are expected to concern themselves with marriage, children, and families. They are not assumed to have responsibilities that force them to work, but rather, are expected to be supported by a father and then a husband.

The third subtheme lists the supportive factors for human development. Almost all interviewees conveyed that Qatar's dynamic leadership supports initiatives to achieve their vision and accelerate development. This supportive factor is also reflected in countries that have experienced a structural change in their economy from a conventional state to one based on knowledge and human potential. These countries have experienced leadership that made timely decisions, reflecting a dynamic and efficient character [82]. Another supportive factor is the Qatar infrastructure for banking and transportation systems, which are considered important for development. One additional supportive factor for developing human capacity is the possibility of collaboration at research and development centers within the country [83]. The Qatar Science Technology Park (QSTP), Qatar University (QU), Hamad Bin Khalifa University (HBKU), and similar institutions must invest further in student development.

The fourth subtheme concerns needs and ideas for what human capacity should look like and what type of structure and skills it must adopt. Many respondents believed that society offers no freedom to participate in progress, intimating that their role in development seemed unknown or unvalued. Issues with workplace demand represent another consideration, as there is a lack of clear requirements regarding what the workplace requires and what universities should produce. This gap was explicitly noted in multiple interviews. Currently, university graduates are typically not recruited by suitable workplaces. Most academics interviewed observed that while many learners are being produced, they are not being aimed at any specified target. Similarly, interviewees from the public and private sectors stated that students learn material unnecessary in the workplace.

Several observations in our qualitative study confirmed existing thoughts on the status of human development in Qatar. First, interviewees showed awareness of the massive infrastructure investments that Qatar has undertaken and further iterated, in line with the literature [84,85], the positive relationship between infrastructure investments and

sustainable economic development. Second, respondents noted that a nation's competitiveness requires a knowledge economy focused on education [86]. However, Qatar falls short in educational outcomes. Third, respondents felt that Qatari men lack incentives to pursue higher education and upgrade their qualifications and that women must be more empowered in their professional growth to contribute further to society and the economy.

### 4.3. Theme 3: Human Development Model for Qatar

In the latter stage of the interview process, participants were asked what changes, policies, and initiatives should be introduced in Qatar to establish a human capacity model and strategy that supports QNV and aligns with SDGs. This theme provides further analysis for RQ2 of this research. First, Qatar must attract highly-skilled individuals to develop its economy and society. To contribute in a measurable way, highly-skilled expatriates must be offered residency, including a range of incentives and the fundamental rights enjoyed by Qatari citizens. Second, a social contract between the state and its citizens must be established. This social contract will define the rights of citizens and inspire them to work toward a common goal. Some participants indicated that general attitudes and mindsets of Qataris must be modified. They pointed out that the education system is an ideal starting place for changes to occur. If the paradigm embraced in Qatar changes and the country establishes a strong philosophy and vision, every strategy and policy can link back to this philosophical foundation.

The present education system in Qatar is fragmented, and each level of schooling has a development plan that is not integrated into a plan for the entire sector [87]. It is vital that universities, schools, and industries set joint goals and strategies and coordinate their aims and activities [88]. Many interviewees noted the importance of encouraging women to take on leadership positions and responsibilities. Since Qatar is a young, emerging country, it is essential to ensure that women have culturally and religiously appropriate support in the workplace. Additionally, entrepreneurship is a potential path for women in Qatar, and the government should invest in female entrepreneurs to boost their contribution to the economy. Attention must also be paid to the number of university majors in professional fields such as medicine and engineering who are likely to graduate soon [88]. Comparing this number to estimates of what Qatar will need and the number of positions that will require filling will help determine steps necessary to fill the gap.

#### 5. Discussion

The salient factors from the statistics and the benchmarking study lead us to the following summary: (a) Qatar's hydrocarbon and construction sectors are the dominant contributors to GDP and growth, with a strong influence on the country's demographics and requirements for human resource development; (b) Qatar's expatriate population consists predominantly of men, most from emerging or relatively developing countries and with humble education backgrounds. These workers are mainly employed on contractual terms, providing unskilled or semi-skilled labor to infrastructure projects associated with the upcoming World Cup; (c) After secondary education, Qatari men are frequently offered lucrative jobs by the government, often in the police or military sectors. This process adversely affects their continuing professional development by undermining their motivation to pursue higher education; (d) Qatari women outnumber their male counterparts in tertiary education by a factor of three to one. However, despite this majority, they cannot gain employment or key positions in the male-dominated oil and construction industries. Moreover, sociocultural constraints often make them settle for less demanding urban jobs; (e) Although education statistics in terms of infrastructure, ICT adaptation, student-teacher ratios, and student-school ratios have improved significantly, these improvements are not yet reflected in the metrics surrounding quality of education. To achieve substantial progress, human development and the educational system must be aligned at all levels [89]. To create an innovative, sustainable-friendly environment, education, training, and research must be put in place, along with a political and regulatory climate that favors innovation; furthermore, stakeholders across the education system must contribute to the sustainable progress of development.

We found several associations in our qualitative study that confirmed existing knowledge regarding the status of human development in Qatar. First, interviewees showed awareness of the massive infrastructure investments that Qatar has undertaken, and they also noted the positive relationship between infrastructure investments and sustainable economic development, which resonates with the literature [84,85]. Second, respondents observed that a nation's competitiveness requires a knowledge economy focused on education, but that Qatar falls short in educational outcomes [86]. Third, respondents acknowledged that Qatari men lack incentives to pursue higher education and upgrade their qualifications, while women lacked sufficient empowerment in their professional growth.

The qualitative survey revealed additional insights beyond these shared perspectives. Qataris may lack a sense of belonging, which is key to the feeling of citizenship. This sense of citizenship is among the most important factors in building cohesiveness and connectedness among members of a society. Since this social inclusivity is assumed to offer a pathway for innovation [90] and sustainable development [91], Qatari leadership should reinforce this feeling of citizenship. Furthermore, participants expressed that the process of *Qatarization* should factor in the labor market structure and education system of the country.

# 6. Framework and Recommendations for Qatar Human Development Strategy—Sustainable Development

The human capital strategy developed in 2009 has not been fully and successfully implemented because of continuously changing circumstances: oil crises, blockades by neighboring countries, new entrants to the natural gas market, and the recent COVID-19 pandemic. Thus, a new approach is needed to support the next phase of human development so that Qatar's population is proactively and dynamically aligned with the development goals and needs of a sustainable knowledge economy [92]. The findings from the previous sections and stakeholder feedback have been used to develop this framework, which completes the analysis of RQ2.

### 6.1. Description of the Qatar Human Development Framework (Q-HD FW)

The proposed framework suggests how human development should be promoted to create a progressive and versatile Qatar. QNV2030 outlines what Qatar wants to achieve. The signing of the memorandum on SDGs makes rapid agreement upon all desired goals even more critical. Thus, this framework might be described as a method of effectively promoting human development based on the country's core values, future vision as mentioned in QNV2030, and Qatar's commitment to SDGs. The framework, which aims to develop human capacity, would pave the way forward and undertake all responsibilities for creating a sustainable economy. The proposed framework has four building blocks, along with motivators and expected outcomes (Figure 7). Our model shows that in addition to a skills framework, health, citizenship, and leadership/management also affect the model.



Figure 7. The proposed framework for the Qatar human development strategy.

Each component of the proposed framework follows a pathway toward improving human development; essentially, these components comprise the basic building blocks to improve human quality for the required economy (i.e., knowledge economy) and, consequently, sustainable development. A brief explanation of each component's target follows:

- 1. Health framework: the target of this framework is to develop a healthcare infrastructure that provides affordable, high-caliber healthcare services for all citizens and reduces infant mortality rate. It provides a healthy education and training infrastructure. The intermediate goal of the health framework is to develop institutions that provide affordable, accessible, and quality health services for all.
- 2. Citizenship framework: the target of this framework is to develop educated, talented, and participative citizens that recognize the cultural, religious, and heritage values of society while actively taking part in this transformation. The citizenship framework has an intermediate goal to develop citizens' attitudes toward others and themselves and ensure that they are socially interactive and economically and environmentally participative throughout the developmental stages.
- 3. Skills framework: this is the primary target, which promotes the mastering of skills and lifelong learning through an improved education system to yield economic gains. The intermediate goal is to significantly improve quality of education to address future economic and social demands.
- 4. Leadership and management framework: this framework aims to encourage and facilitate partnerships between industries, universities, K–12 institutions, and the government to promote technology development, research, and innovation. The intermediate goal of this framework is to provide efficient and effective e-services in all aspects of government, business, industry, trade, finance, and education, and to encourage domestic-related technology, ICT, research, and innovation.

The proposed framework advocates for a healthy citizen who is willing to appreciate their culture, society, and religion but also possesses the right capabilities and skills to adapt to changing conditions.

#### 6.2. Recommendations for Qatar's Human Development Strategy

This section introduces and examines the most effective actions Qatar's government could undertake to improve its policies across educational, public, and private sectors. It is worth noting that the recommendations are designed to support Qatar's innovation, research, and knowledge development toward sustainable development. For a successful human development plan, it is imperative that each sector and individual at all levels recognize their role and play an active part in the plan's fulfillment [93]. Positions must be

clearly defined to avoid any ambiguity. The functions have been divided into three main sectors, described below.

# 6.2.1. Education and Higher Education

The education sector is the main contributor to human capital development in Qatar. Previous research has shown that education plays the most important role in health sector improvement, economic growth, and sustainable human growth by increasing workforce productivity, and providing a foundation for a successful democratic government [94]. Therefore, to achieve sustainable development, the education sector must become more adaptable and meet society's changing needs [95]. The education sector must work closely with other workforce sectors to ensure that education focuses on the areas required by the country for aiding its development. The role of the education sector is manifold: it must recognize demands, meet them by aligning itself to critical focus areas, coordinate and continuously access changing dynamics, prepare for unforeseen future requirements, promote the focus sectors through scholarships and incentives, inculcate a sense of tolerance and respect for people from other cultures with different values, and shape citizens of all backgrounds [96]. Our specific recommendations are:

- Building capacity by developing skills and talent through vocational and technical education;
- Improving K–12 education to make students more competitive and successful in reading, mathematics, and science;
- Innovation and entrepreneurship training and providing students and teachers at the K–12 levels with opportunities for experimentation;
- Wider inclusion of male Qatari citizens in the teaching career path through improvement of the current admission and evaluation procedures, and providing role models for increasing the teaching profession's attractiveness;
- Adapting higher education study programs to industry needs and adopting and developing new programs;
- Preparing, promoting, and empowering women and local academics, researchers, and leadership without sacrificing quality and high aspirations.

#### 6.2.2. Government

The state plays a significant role in both shaping the plan and implementing it effectively. The state determines the overall target and direction for the project; thus, state institutions must be an active part of this change at the national and regional levels [97]. The state must also determine the distribution of responsibilities and develop committees to evaluate performance. Coordination among different state institutions, both internally and externally, would also fall to the state. In this regard, the government can:

- Promote entrepreneurship, particularly for female entrepreneurs and investors;
- Provide targeted naturalization and selective permanent residency for highly-skilled ex-pats.

#### 6.2.3. Industry/Business/Private Sector

Significant contributions in the health and education sectors have primarily been made by the state. However, favorable government policies have also attracted the private sector. The private sector must contribute to the shift toward human development by understanding and applying government guidelines and coordinating with authorities to effectively follow the action plan [97]. Private sector investments in critical areas would pave the way for successful implementation of the proposed framework through the establishment of industry–university–government partnerships in education, training, and research. Business incubators are further recommended to diffuse knowledge and thus contribute to economic growth [98].

#### 7. Conclusions, Limitations, and Future Work

Sustainable development requires meeting current needs without compromising future generations' resources, and human capital forms a crucial component [99–101]. The importance of human capital was recognized by the United Nations when they established the Sustainable Development Goals (SDGs) in 2015. Of these goals, the most relevant to human capital is SDG4, which aims to "ensure inclusive and equitable quality education and promote lifelong opportunities for all." As the rest of the world moves steadily toward embracing new technologies as a source of long-term growth, it has become clear that Qatar's path to sustainable development and diversification requires replacing the hydrocarbon-led economic model with a strategy for developing a knowledge-based economy in which the ITC sector and other high-value industries can thrive. This study aimed to develop a comprehensive framework to examine and assess the human development strategies of Qatar by exploring the developments introduced in multiple areas, including education, technology, economic diversification, and Qatarization. This exercise is vital, as Qatar faces considerable challenges in creating innovation building blocks and promoting innovation performance. We discuss the pathway by which Qatar can enhance understanding of human development strategies and transition from a resource-led growth model to a knowledge-based economy. More specifically, we assess the effectiveness of current education reforms and recommend strategies to address emerging policy challenges in Qatar. Such strategies could enable Qatar to achieve the SDGs and meet the needs of knowledge-intensive industries. The proposed framework can serve as a way forward for Qatar and other countries with similar educational and demographic dynamics where the framework could be adapted to various requirements. Research is lacking concerning countries in the Middle East; such a framework could thus present a stepping stone for Qatar, by which it could incorporate teachings from other countries that embarked upon similar transitions and achieved knowledge-based economies.

To gain better insight into these research aims, a qualitative study was conducted to gather the opinions and perspectives of various participants from the public, private, and academic sectors to explore the status of human development in Qatar and develop a better understanding of adopted strategies, existing challenges and issues, and contributions of different sectors to the development process. Furthermore, participant recommendations and suggestions for improving the human capacity required to support a knowledge-based economy in Qatar were gathered and reviewed. Interviews were analyzed using a thematic approach to identify significant elements related to themes arising from the current infrastructure in Qatar, and the affecting and transformation factors that are assisting or impeding the creation of a successful human development model for Qatar.

Based on literature review, observations, and interview results, this study proposes a human development framework for a progressive and versatile Qatar. The primary driver for this framework is the development of human capacity, which would pave the way forward and undertake responsibilities for sustainable development. The framework has four components: health, citizens, skills, and leadership and management. These four components can be described as a method of effectively promoting human development based on Qatar's core values, commitment to SDGs, and the country's future vision, as delineated in QNV2030.

Some key limitations in implementing the framework are related to the country's population, geography, and climate. The population growth rate is low and is expected to decline further. Hence, decision-makers must incorporate an export-based niche market to attract foreign investment. Furthermore, the geographical conditions and political factors of surrounding countries represent another crucial area of focus for decision-makers, since an effective framework for developing policies cannot be created without incorporating these risks. A proactive stance is needed to develop progressive human capacity policies and effective practices to increase retention rates, including transparent citizenship, long-term residency, and careful human resources practices. In addition, equal opportunity employment, business establishment, and sophisticated technological hubs are needed for

Qatar to transition from a rentier state to a knowledge-based economy. The government must develop specific policies and programs to diligently implement and continuously improve mechanisms that focus on such themes as "plan ahead and carefully," "efficiency always first," "first time right approach," "diligent cost-control," "speed," and "quality" for all operations of government, industry, and private sector investments and institutions. This is particularly important for educational institutions, as they shape the minds and hearts of future generations not only by teaching but also by doing it the right way. Only when this is achieved can the suggested framework help make the sustainability targets more meaningful and accessible while keeping Qatar's native population at the core of all

Author Contributions: Conceptualization, B.H.M., M.D., M.b.S.A.-S., and M.K.; data curation, B.H.M., M.D., and M.K.; formal analysis, B.H.M., M.D., and M.K.; funding acquisition, M.D.; investigation, B.H.M., M.D., M.b.S.A.-S., and M.K.; methodology, B.H.M., M.D., and M.K.; project administration, B.H.M., M.D., M.b.S.A.-S., and M.K.; resources, B.H.M., M.D., and M.K.; software, B.H.M., M.D., and M.K.; supervision, M.D., M.b.S.A.-S., and M.K.; validation, M.D. and M.K.; visualization, B.H.M., M.D., and M.K.; writing—original draft, B.H.M.; writing—review and editing, M.D. and M.K. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

**Institutional Review Board Statement:** The study was conducted according to the guidelines of the Declaration of Helsinki and approved by the Institutional Review Board of Qatar Biomedical Research Institute (protocol code: QBRI-IRB-2020-07-025, approval date: 27 July 2020).

Informed Consent Statement: Informed consent was obtained from all subjects involved in the study.

Data Availability Statement: Not applicable.

Conflicts of Interest: The authors declare no conflict of interest.

# Appendix A

human development.

A.1. Interview Guide: Below Are the Main Aspects/Areas of Focus in the Interviews

Qatar's current status pertaining to the knowledge economy;

Strategies adopted by Qatar for moving human capital to the knowledge economy; Challenges faced in moving to a knowledge economy;

Qatar's future with regard to the knowledge economy;

Actions/sectors Qatar should take and invest in to move to the knowledge economy; Strategies and steps taken for successful human resource development (what, who, and how);

Suggestions/recommendations to improve the knowledge economy in Qatar.

# A.2. Interview Questions

Based on your background, experience, and knowledge, how do you think Qatar arrived at today's economic development status from the 1970s? What were the motivations, drivers, and challenges?

How do you think the human development aspect of the above changes played out? Equally developed? Stayed behind? Why?

What has played a critical role to allow human development to follow an equal trend or stay behind the economic development of Qatar since the 1970s?

In regard to the population in Qatar and Qatari population % today, how do you see their participation % in the Qatari workforce (in term of skills and education level)?

Do you think that the above % are sustainable? Why?

In terms of economic structure and development, please compare today's and 2030's goals? Share of LNG revenues? Share of LNG in GDP? What are the gaps and why?

In terms of demographic structure and development, please compare today's and 2030's goals? What are the gaps and why?

In terms of human capacity (education, knowledge, talent), structure, and development, please compare today's and 2030's goals? What are the gaps and why?

In terms of workforce/labor (skill sets, blue-collar vs. white-collar), structure, and development, please compare today's and 2030's goals? What are the gaps and why?

What should be the ideal economy sectors in 2030 for sustainable economy and development? Which sectors/industries should Qatar invest in and grow? Why?

How can ideal economic diversification (sectors, industries, businesses; talents, skills; financing) be designed? What are the factors and conditions for it?

What do you think Qatar should do now to achieve a sustainable economy and diversification in 2030?

What do you think the 2030 Human Capacity needs are? Which skills, knowledge sets, talents? Why?

How can the ideal human capital/capacity/development (education, health, population growth, Qatari vs. non-Qatari, male vs. female, high-skilled vs. low-skilled) be designed? What are the factors and conditions for it?

What do you think Qatar should do now to achieve the right human capacity development in 2030?

#### References

- 1. Kraay, A. The World Bank human capital index: A guide. World Bank Res. Obs. 2019, 34, 1–33. [CrossRef]
- Schröder, J. Decoupling of Labour Productivity Growth from Median Wage Growth in Central and Eastern Europe. The Vienna Institute for International Economic Studies: Vienna, Austria, 2020. Available online: <a href="https://wiiw.ac.at/decoupling-of-labour-productivity-growth-from-median-wage-growth-in-central-and-eastern-europe-dlp-5356.pdf">https://wiiw.ac.at/decoupling-of-labourproductivity-growth-from-median-wage-growth-in-central-and-eastern-europe-dlp-5356.pdf</a> (accessed on 21 January 2022).
- Barnett, M.L.; Darnall, N.; Husted, B.W. Sustainability strategy in constrained economic times. Long Range Plan. 2015, 48, 63–68. [CrossRef]
- Hertog, S. What would the Saudi economy have to look like to be post rentier? POMEPS Stud. 2019, 12, 29–33.
- 5. Young, R.F.; Lieberknecht, K. From smart cities to wise cities: Ecological wisdom as a basis for sustainable urban development. *J. Environ. Plan. Manag.* **2019**, *62*, 1675–1692. [CrossRef]
- 6. Gollin, D.; Jedwab, R.; Vollrath, D. Urbanization with and without industrialization. J. Econ. Growth 2016, 21, 35–70. [CrossRef]
- Venables, A.J. Using natural resources for development: Why has it proven so difficult? J. Econ. Perspect. 2016, 30, 161–184. [CrossRef]
- 8. Gylfason, T. Natural resources, education, and economic development. Eur. Econ. Rev. 2001, 45, 847–859. [CrossRef]
- Al-Suwaidi, J.S. Towards a Strategy to build Administrative Capacity in Light of human Development for Qatar National Vision 2030. Ph.D. Dissertation, University of Bolton, Bolton, UK, 2018.
- 10. Weber, A.S. *E-Learning Middle East North Africa (MENA) Region;* Weber, A.S., Hamlaoui, S., Eds.; Springer: Berlin/Heidelberg, Germany, 2018; pp. 333–354.
- 11. Doz, Y.; Santos, J.; Williamson, P. From Global to Metanational: How Companies Win in the Knowledge Economy; Harvard Business School Press: Boston, MA, USA, 2001.
- Phale, K.; Fanglin, L.; Adjei Mensah, I.; Omari-Sasu, A.Y.; Musah, M. Knowledge-based economy capacity building for developing countries: A panel analysis in Southern African development community. *Sustainability* 2021, 13, 2890. [CrossRef]
- 13. Koolivand, A.; Salehi, M.; Arabzadeh, M.; Ghodrati, H. The relationship between knowledge-based economy and fraudulent financial reporting. *J. Facil. Manag.* 2021, *8*, 116–127. [CrossRef]
- 14. Asongu, S.; Nwachukwu, J. The incremental effect of education on corruption: Evidence of synergy from lifelong learning. *Econ. Bull.* **2015**, *35*, 2288–2308. [CrossRef]
- 15. Carnoy, M.; Castells, M. Globalization, the knowledge society, and the Network State: Poulantzas at the millennium. *Glob. Netw.* **2001**, *1*, 1–18. [CrossRef]
- 16. Gremm, J.; Barth, J.; Fietkiewicz, K.J.; Stock, W.G. *Transitioning towards a Knowledge Society: Qatar as a Case Study*; Springer: Cham, Switzerland, 2018.
- Bafarasat, A.Z.; Oliveira, E. Prospects of a transition to the knowledge economy in Saudi Arabia and Qatar: A critical reflection through the lens of spatial embeddedness and evolutionary governance theory. *Futures* 2021, 129, 102731. [CrossRef]
- Castells-Quintana, D.; Royuela, V.; Thiel, F. Inequality and sustainable development: Insights from an analysis of the human development index. *Sustain. Dev.* 2019, 27, 448–460. [CrossRef]
- 19. Kaplanidou, K.K.; Al Emadi, A.; Sagas, M.; Diop, A.; Fritz, G. Business legacy planning for mega events: The case of the 2022 World Cup in Qatar. *J. Bus. Res.* **2016**, *69*, 4103–4111. [CrossRef]
- Absalyamov, T. The influence of cultural and sport mega-events on sustainable development of the city. *Procedia-Soc. Behav. Sci.* 2015, 188, 197–201. [CrossRef]

- 21. Shamu, G.G. The Developmental Impacts of FIFA World Cups on BRICS Nations. Master's Thesis, University of the Witwatersrand, Johannesburg, South Africa, 2016.
- 22. De Aragao, M.M. *Economic Impacts of the FIFA World Cup in Developing Countries;* Honors Theses 2609; Western Michigan University: Kalamazoo, MI, USA, 2015.
- 23. Sim, L.-C. Low-carbon energy in the Gulf: Upending the rentier state? Energy Res. Soc. Sci. 2020, 70, 101752. [CrossRef]
- 24. Jin, H.; Qian, X.; Chin, T.; Zhang, H. A global assessment of sustainable development based on modification of the human development index via the entropy method. *Sustainability* **2020**, *12*, 3251. [CrossRef]
- 25. Metcalfe, B.D. Gender and human resource management in the Middle East. *Int. J. Hum. Resour. Manag.* 2007, 18, 54–74. [CrossRef]
- Winterton, J.; Cafferkey, K. Revisiting human capital theory: Progress and prospects. In *Introduction to Theories of Human Resources and Employment Relations*; Townsend, K., Cafferkey, K., McDermott, A.M., Dundon, T., Eds.; Edward Elgar Publishing: Cheltenham, UK, 2019; Chapter 15; pp. 218–234.
- Mohamed, B.H.; Ari, I.; Al-sada, M.S.; Koç, M. Strategizing Human Development for a Country in Transition from a Resource-Based to a Knowledge-Based Economy. *Sustainability* 2021, 13, 13750. [CrossRef]
- Chansarn, S. The evaluation of the sustainable human development: A cross-country analysis employing slack-based DEA. Procedia Environ. Sci. 2014, 20, 3–11. [CrossRef]
- 29. Sohaib, O.; Fahad Assad, A.O. Enhancing innovative capability and sustainability of saudi firms. Sustainability 2016, 8, 1229.
- 30. Sillitoe, P.; Alshawi, A.A. Conservation and sustainable development: The Qatari and Gulf Region experience. In *Sustainable Development: An Appraisal from the Gulf Region*; Sillitoe, P., Ed.; Berghahn Books: New York, NY, USA, 2014; Chapter 10; pp. 221–269.
- 31. Rutledge, E.; Al Shamsi, F.; Bassioni, Y.; Al Sheikh, H. Women, labour market nationalization policies and human resource development in the Arab Gulf states. *Hum. Resour. Dev. Int.* 2011, 14, 183–198. [CrossRef]
- OPEC Qatar Natural Gas Production: OPEC: Marketed Production. Available online: https://www.ceicdata.com/en/indicator/ qatar/natural-gas-production-opec-marketed-production (accessed on 1 March 2022).
- 33. Sahin, E.S.; Bayram, I.S.; Koc, M. Demand side management opportunities, framework, and implications for sustainable development in resource-rich countries: Case study Qatar. J. Clean. Prod. 2019, 241, 118332. [CrossRef]
- Ben Hassen, T. Entrepreneurship, ICT and Innovation: State of Qatar Transformation to a Knowledge-Based Economy. In *Qatar: Political, Economic and Social Issues*; Nova Science Publishers: Hauppauge, NY, USA, 2019; Chapter 7; pp. 193–208.
- 35. Meza Talavera, A.; Al-Ghamdi, S.G.; Koç, M. Sustainability in mega-events: Beyond Qatar 2022. *Sustainability* **2019**, *11*, 6407. [CrossRef]
- OnlineQatar. Qatar Population and Expat Nationalities. Available online: https://www.onlineqatar.com/visiting/touristinformation/qatar-population-and-expat-nationalities (accessed on 26 January 2022).
- OECD. PISA 2018 Results (Volume III); OECD: Paris, France, 2019; Available online: https://www.oecd.org/pisa/Combined\_ Executive\_Summaries\_PISA\_2018.pdf (accessed on 21 January 2022).
- Bergerhoff, J.; Borghans, L.; Seegers, P.K.; Van Veen, T. International education and economic growth. *IZA J. Eur. Labor Stud.* 2013, 2, 1–13. [CrossRef]
- Brockmann, C.; Roztocki, N. The six pillars of knowledge economics. In Proceedings of the 50th Hawaii International Conference on System Sciences, HICSS, Hawaii, HI, USA, 4–7 January 2017; University of Hawaii: Manoa, HI, USA, 2017.
- 40. Planning and Statistics Authority in Qatar. Woman and Man in the State of Qatar: A Statistical Portrait. Qatar Plan. Stat. Auth. 2018. Available online: https://www.psa.gov.qa/en/Pages/default.aspx (accessed on 22 January 2022).
- 41. Fadlelmula, F.; Koc, M. Overall review of education system in Qatar; Lambert Academic Publishing: Chisinau, Moldova, 2016.
- 42. Said, Z. Science education reform in Qatar: Progress and challenges. *Eurasia J. Math. Sci. Technol. Educ.* 2016, 12, 2253–2265. [CrossRef]
- 43. Card, D.; Krueger, A.B. Does school quality matter? Returns to education and the characteristics of public schools in the United States. *J. Polit. Econ.* **1992**, 100, 1–40. [CrossRef]
- 44. Govind, N. International Journal of Social Science and Economic Research. Int. J. Soc. Sci. Econ. Res. 2019, 4, 2792–2806.
- 45. Silverman, D. Qualitative Research: Issues of Theory, Method and Practice, 3rd ed.; SAGE: Los Angeles, CA, USA, 2011.
- 46. Ali, M.H. The Influence of International Tests on Arabic Educational Policies Through Examining the Results of Arab Countries in PISA. J. Xi'an Univ. Archit. Technol. 2020, XII, 257–270.
- 47. OECD. What makes urban schools different? PISA Focus 2013, 28, 1-4.
- Parcero, O.J.; Ryan, J.C. Becoming a knowledge economy: The case of Qatar, UAE, and 17 benchmark countries. J. Knowl. Econ. 2017, 8, 1146–1173. [CrossRef]
- Al-Kuwari, M.M.; Al-Fagih, L.; Koç, M. Asking the right questions for sustainable development goals: Performance assessment approaches for the Qatar education system. *Sustainability* 2021, 13, 3883. [CrossRef]
- Al-Kuwari, M.M.; Du, X.; Koç, M. Performance assessment in education for sustainable development: A case study of the Qatar education system. *Prospects* 2021. ahead-of-print. [CrossRef]
- European Youth Forum. Policy Paper on Sustainable Development. Available online: https://tools.youthforum.org/policylibrary/wp-content/uploads/2021/04/0047-18\_COMEMI-18\_PP\_SustainableDevelopment\_FINAL.docx\_.pdf (accessed on 21 January 2022).

- 52. Kioupi, V.; Voulvoulis, N. Education for sustainable development: A systemic framework for connecting the SDGs to educational outcomes. *Sustainability* **2019**, *11*, 6104. [CrossRef]
- World Bank. The Five-Year Socio-Economic Development Plan From 2016–2020. World Bank: Hanoi, Vietnam. 2016. Available online: https://pubdocs.worldbank.org/en/839361477533488479/pdf/Vietnam-SEDP-2016-2020.pdf (accessed on 21 January 2022).
- 54. Permanand, G.; Lindahl, A.K.; Røttingen, J.A. Knowledge brokering in Norway: Bringing rigour and transparency to policy inputs. In *Bridging the worlds of research and policy in European health systems;* Lavis, J.N., Catallo, C., Eds.; WHO Regional Office for Europe on behalf of the European Observatory on Health Systems and Policies: Copenhagen, Denmark, 2013; Chapter 8; pp. 1–29.
- Fitjar, R.D.; Timmermans, B. Knowledge bases and relatedness: A study of labour mobility in Norwegian regions. In *New Avenues for Regional Innovation Systems-Theoretical Advances, Empirical Cases and Policy Lessons*; Springer: Berlin/Heidelberg, Germany, 2018; pp. 149–171.
- 56. Csizmazia, R. Singapore Systematically Running Globalization Efforts. Int. J. Knowl. Innov. Bus. 2013, 3, 1–22.
- 57. United Nations. *Human Development Report 2020. The Next Frontier: Human Development and the Anthropocene;* United Nations: New York, NY, USA, 2020.
- UN Human Development Index (HDI) by Country. Available online: https://worldpopulationreview.com/country-rankings/ hdi-by-country (accessed on 21 January 2022).
- UNDP Human Development Report: Education Index. Available online: http://hdr.undp.org/en/indicators/103706# (accessed on 21 January 2021).
- UNDP. Human Development Reports. Available online: https://hdr.undp.org/en/content/latest-human-development-indexranking (accessed on 20 December 2021).
- 61. Jolo, H.A. Human Capital Formation: The Case of Qatar. Stud. Bus. Econ. 2005, 7, 67–93. [CrossRef]
- 62. WIPO. *Global Innovation Index 2021: Tracking Innovation through the COVID-19 Crisis;* World Intel-lectual Property Organization: Geneva, Switzerland, 2021.
- Nawaz, W.; Koç, M. Comparison Between the IUGP Settings and Global Innovation Index of Qatar, United States, Norway, and Singapore. In *Industry, University and Government Partnerships for the Sustainable Development of Knowledge-Based Society;* Springer: Berlin/Heidelberg, Germany, 2020; pp. 107–131.
- 64. Al-Thani, W.A.; Ari, I.; Koç, M. Education as a Critical Factor of Sustainability: Case Study in Qatar from the Teachers' Development Perspective. *Sustainability* **2021**, *13*, 1525. [CrossRef]
- 65. Britten, N. Qualitative research: Qualitative interviews in medical research. BMJ 1995, 311, 251–253. [CrossRef]
- 66. Puig, G.V.; Al-Haddab, B. The protection of minority shareholders in the gulf cooperation council. *J. Corp. Law Stud.* **2013**, *13*, 123–149. [CrossRef]
- 67. Jenner, B.; Flick, U.; von Kardoff, E.; Steinke, I. A Companion to Qualitative Research; Sage: London, UK, 2004.
- 68. Kallio, H.; Pietilä, A.-M.; Johnson, M.; Kangasniemi, M. Systematic methodological review: Developing a framework for a qualitative semi-structured interview guide. *J. Adv. Nurs.* **2016**, *72*, 2954–2965. [CrossRef] [PubMed]
- 69. Kumi, E.; Yeboah, T.; Kumi, Y.A. Private sector participation in advancing the sustainable development goals (SDGs) in Ghana: Experiences from the mining and telecommunications sectors. *Extr. Ind. Soc.* **2020**, *7*, 181–190. [CrossRef]
- 70. Braun, V.; Clarke, V. Using thematic analysis in psychology. Qual. Res. Psychol. 2006, 3, 77–101. [CrossRef]
- Nowell, L.S.; Norris, J.M.; White, D.E.; Moules, N.J. Thematic Analysis: Striving to Meet the Trustworthiness Criteria. Int. J. Qual. Methods 2017, 16, 1–13. [CrossRef]
- 72. Peel, K.L. Beginner's Guide To Applied Educational Research Using Thematic Analysis. *Pract. Assess. Res. Eval.* **2020**, 25, 1–16.
- 73. Ferri, N. United nations general assembly. Int. J. Mar. Coast. Law 2010, 25, 271–287. [CrossRef]
- 74. Huston, A.C.; Bentley, A.C. Human development in societal context. Annu. Rev. Psychol. 2010, 61, 411–437. [CrossRef]
- 75. Valsiner, J. Culture and Human Development; Sage: London, UK, 2000.
- Venkatraja, B.; Indira, M. Role of education in social development: An empirical analysis. *Madhya Pradesh J. Soc. Sci.* 2011, 16, 1–11.
- 77. Berrebi, C.; Martorell, F.; Tanner, J.C. Qatar's labor markets at a crucial crossroad. Middle East J. 2009, 63, 421–442. [CrossRef]
- 78. Sulaiman, N.F.; Manochehri, N.-N.; Al-Esmail, R.A. Level of total quality management adoption in Qatari educational institutions: Private and semi-government sector. *J. Educ. Bus.* **2013**, *88*, 76–87. [CrossRef]
- 79. Gonzalez, G.C.; Karoly, L.A.; Constant, L.; Salem, H.; Goldman, C.A. Facing Human Capital Challenges of the 21st Century: Education and Labor Market Initiatives in Lebanon, Oman, Qatar, and the United Arab Emirates; Rand Corporation: Santa Monica, CA, USA, 2008.
- Guarino, C.M.; Tanner, J.C. Adequacy, accountability, autonomy and equity in a Middle Eastern school reform: The case of Qatar. Int. Rev. Educ. 2012, 58, 221–245. [CrossRef]
- Disli, M.; Yilmaz, M.K.; Mohamed, F.F.M. Board characteristics and sustainability performance: Empirical evidence from emerging markets. Sustain. Account. Manag. Policy J. 2022. ahead-of-print. [CrossRef]
- Shterev, N.; Stoyanova, T.; Parushev, D. Knowledge Based Economy a Great Challenge to Leadership Models in Developing Countries (Bulgarian Case). In Proceedings of the 2019 IEEE Jordan International Joint Conference on Electrical Engineering and Information Technology (JEEIT), Amman, Jordan, 9–11 April 2019; pp. 817–821.

- 83. Wong, T.; Dutfield, G. Intellectual Property and Human Development: Current Trends and Future Scenarios; Cambridge University Press: Cambridge, UK, 2010.
- 84. Esfahani, H.S.; Ramírez, M.T. Institutions, infrastructure, and economic growth. J. Dev. Econ. 2003, 70, 443–477. [CrossRef]
- 85. Adshead, D.; Thacker, S.; Fuldauer, L.I.; Hall, J.W. Delivering on the Sustainable Development Goals through long-term infrastructure planning. *Glob. Environ. Chang.* **2019**, *59*, 101975. [CrossRef]
- Jednak, S.; Kragulj, D. Achieving sustainable development and knowledge-based economy in Serbia. Manag. J. Sustain. Bus. Manag. Solut. Emerg. Econ. 2015, 20, 1–12.
- 87. Nasser, R. Qatar's educational reform past and future: Challenges in teacher development. *Open Rev. Educ. Res.* 2017, 4, 1–19. [CrossRef]
- Al-Hail, M.A.; Al-Fagih, L.; Koç, M. Partnering for Sustainability: Parent-Teacher-School (PTS) Interactions in the Qatar Education System. Sustainability 2021, 13, 6639. [CrossRef]
- 89. Urmetzer, S.; Lask, J.; Vargas-Carpintero, R.; Pyka, A. Learning to change: Transformative knowledge for building a sustainable bioeconomy. *Ecol. Econ.* **2020**, *167*, 106435. [CrossRef]
- 90. Salehi, M.; Fahimi, M.A.; Zimon, G.; Homayoun, S. The effect of knowledge management on intellectual capital, social capital, and firm innovation. *J. Facil. Manag.* 2021. ahead-of-print. [CrossRef]
- 91. Dobson, A. Environmental citizenship: Towards sustainable development. Sustain. Dev. 2007, 15, 276–285. [CrossRef]
- Al Meraikhi, H. Adoption of Smart and Sustainable Strategies in the State of Qatar. PhD Dissertation, University of Wolverhampton, Wolverhampton, UK, 2021.
- 93. Ranis, G.; Stewart, F. Strategies for success in human development. J. Hum. Dev. 2000, 1, 49-69. [CrossRef]
- 94. Aloui, L.; Shams Eldin, A.Y. Socio-emotional competencies, socio-economic factors, and the employability process of female graduates in the Kingdom of Saudi Arabia. *Int. J. Eng. Bus. Manag.* **2020**, *12*, 1–14. [CrossRef]
- 95. Al-Mulla, S.; Ari, I.; Koç, M. Social media for sustainability education: Gaining knowledge and skills into actions for sustainable living. *Int. J. Sustain. Dev. World Ecol.* 2022. ahead-of-print. [CrossRef]
- 96. Randers, J.; Rockström, J.; Stoknes, E.; Golüke, U.; Collste, D.; Cornell, S.E. Transformation is feasible: How to achieve the Sustainable Development Goals within Planetary Boundaries. *Stock. Resil. Cent.* **2018**, *60*, 24–67.
- World Bank. Global Monitoring Report 2009: A Development Emergency; World Bank: Washington, DC, USA, 2009; Available online: https://documents1.worldbank.org/curated/en/703621468339684203/pdf/484570PUB0glob101Official0Use0Only1.pdf (accessed on 22 January 2022).
- Binsawad, M.; Sohaib, O.; Hawryszkiewycz, I. Factors impacting technology business incubator performance. Int. J. Innov. Manag. 2019, 23, 1950007. [CrossRef]
- 99. Al-Shaiba, A.; Al-Ghamdi, S.G.; Koç, M. Measuring efficiency levels in Qatari organizations and causes of inefficiencies. *Int. J. Eng. Bus. Manag.* 2020, 12, 1–18. [CrossRef]
- Aysan, A.F.; Bergigui, F.; Disli, M. Using Blockchain-Enabled Solutions as SDG Accelerators in the International Development Space. Sustainability 2021, 13, 4025. [CrossRef]
- Aysan, A.F.; Bergigui, F.; Disli, M. Blockchain-based solutions in achieving SDGs after COVID-19. J. Open Innov. Technol. Mark. Complex. 2021, 7, 151. [CrossRef]